Author’s response to reviews

Title: A qualitative analysis of information sharing for children with medical complexity within and across health care organizations

Authors:

Laura Quigley (laura.quigley@mail.utoronto.ca)
Ashley Lacombe-Duncan (ashley.lacombe-duncan@sickkids.ca)
Sherri Adams (sherri.adams@sickkids.ca)
Charlotte Moore (charlotte.moorehepburn@sickkids.ca)
Eyal Cohen (eyal.cohen@sickkids.ca)

Version: 2 Date: 31 May 2013

Author’s response to reviews: see over
Re: A Qualitative Analysis of Information Sharing for Children with Medical Complexity Within and Across Health Care Organizations

To the Editorial Board of BMC Health Services Research,

Please find the responses to the editorial concerns below:

1. Name of ethics committee

   Please update your ethics statement to include the name of the ethics committee that approved your study.

   - The manuscript has been updated indicating that the names of the ethics committees which approved the study were: The Hospital for Sick Children Research Ethics Board, the Bloorview Research Institute Research Ethics Board, and the Joint Research Ethics Board

2. Please adhere to the RATS guidelines and format your manuscript according to our guidelines: (http://www.biomedcentral.com/bmchealthservres/authors/instructions/researcharticle#preparing-main-manuscript).

   - The title was reformatted so that only the first letter of the first word was capitalized
   - The section ‘Abbreviations’ was moved from the location directly following the title page, and is now found directly following the Conclusions
   - The section Declarations was retitled ‘Competing Interests’ and moved from the location directly following the title page, and is now found following the new location of the Abbreviations section
   - The section ‘Contributions’ was moved from the location following the title page, and is now found following the new location of the Declarations section
   - The section ‘Acknowledgements’ was added and placed preceding the References section

Please let me know if you require any other information or changes to the manuscript. Thank you!

Laura Quigley